Author’s response to reviews

Title: Evaluation of the undergraduate Family Medicine programme of Faculty of Medicine, University of Kelaniya: Quantitative and qualitative student feedback

Authors:

Dinusha Perera (dinusha@kln.ac.lk)
Supun Withana (sswithana@kln.ac.lk)
Kumara Mendis (kmendis@kln.ac.lk)
Thilanka Kasunjith (thilankakasunjith@gmail.com)
Tharanga Jayathilaka (tharangasjay@yahoo.com)
Sachie Wickramasuriya (sachiewathsala@gmail.com)

Version: 1 Date: 25 Apr 2019

Author’s response to reviews:

25.04.2019

Liam Messin
Editor
BMC Medical Education

Dear Editor,

MEED-D-18-00835- Evaluation of the undergraduate Family Medicine clinical appointment of Faculty of Medicine, University of Kelaniya: Quantitative and qualitative feedback

I am submitting the revisions requested.

1. Title page added with names and emails of all authors, institution and country, corresponding authors postal address

2. Heading of abstract added

3. Declaration heading added and all sections included.

4. Ethics approval changed to ethics and consent to participate
thank you for the time of all staff concerned. Hope all is in order now.

with regards

D.P.Perera